

The state of the art of innovation-driven business models in the financial services industry

Citation for published version (APA): Lüftenegger, E. R., Angelov, S. A., Linden, van der, E., & Grefen, P. W. P. J. (2010). *The state of the art of innovation-driven business models in the financial services industry.* (BETA publicatie : working papers; Vol. 310). Technische Universiteit Eindhoven.

Document status and date: Published: 01/01/2010

Document Version:

Publisher's PDF, also known as Version of Record (includes final page, issue and volume numbers)

Please check the document version of this publication:

• A submitted manuscript is the version of the article upon submission and before peer-review. There can be important differences between the submitted version and the official published version of record. People interested in the research are advised to contact the author for the final version of the publication, or visit the DOI to the publisher's website.

• The final author version and the galley proof are versions of the publication after peer review.

• The final published version features the final layout of the paper including the volume, issue and page numbers.

Link to publication

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- · Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
 You may freely distribute the URL identifying the publication in the public portal.

If the publication is distributed under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license above, please follow below link for the End User Agreement:

www.tue.nl/taverne

Take down policy

If you believe that this document breaches copyright please contact us at:

openaccess@tue.nl

providing details and we will investigate your claim.

The state of the art of innovation-driven business models in the financial services industry

E. Lüftenegger¹, S. Angelov¹, E. van der Linden², and P. Grefen¹

¹School of Industrial Engineering, Eindhoven University of Technology ²De Lage Landen Financial Services

Eindhoven, The Netherlands, March, 2010

Abstract

Emerging innovation-driven business models are changing the financial services landscape. Most companies are using innovation to sustain their business models. However, new entrants into the financial services market innovate in a way that disrupts the industry. Typically, directions for innovation initiatives in financial services are absent. In this report, we present a structured method to analyze innovation initiatives and their impact on the financial services industry. Our method is based on innovation and business model frameworks that let us analyze business models driven by different kinds of innovations. We apply our method to emerging innovation-driven business models providing an overview of the financial services industry. Companies in financial services can use this report as an overview of the state of the art and as a guiding tool for their innovation initiatives. We contribute to the innovation and business models research fields, presenting a unified method to analyze business models driven by innovation in financial services.

Contents

I	Introduction	2
2	Overview of the financial services landscape2.1Market forces.2.2Industry forces.2.3Key trends.2.4Macroeconomic forces.	3 3 4 5 6
3	Financial services innovation framework3.1Innovation dimensions3.2The innovation pentagram for the financial services industry	7 7 10
4	Business model frameworks4.1 The business model canvas4.2 Business model for e-business4.3 Summary of this section	14 14 16 18
_		
5	A method to analyze innovation-driven business models in the finan cial services industry	- 19
6	A method to analyze innovation-driven business models in the financial services industry Emerging innovation-driven business models in the financial industry 6.1 Peer-to-Peer lending	- 19 - 21 23 26 29 32 34 37
5 6 7	A method to analyze innovation-driven business models in the financial services industry Emerging innovation-driven business models in the financial industry 6.1 Peer-to-Peer lending	- 19 211 23 26 29 32 34 37 38

1 Introduction

The financial services sector represents about 8% of the gross domestic product (GDP) and 4% of employment in the service economy (70% of the total GDP) of the Organization for Economic Cooperation and Development (OECD) countries [1]. Financial services include a broad range of businesses. They encompass businesses such as banks, consumer finance, stock brokerage, asset management, credit card companies and insurance companies [2].

The recent financial industry crisis caused by the meltdown in the U.S. market has generated huge losses in this sector leading to a new economic scenario called "the Great Disruption" [3] [4] [2]. The new economic environment requires financial services institutions to innovate, redefine their business models and restore the trust of their clients in order to improve the industry's health and long-term growth [5].

Innovations and business models are related. Financial services companies are using innovations as a driver to support or change their business models. Innovations are used to support the current business model of a company through products, customer experiences, markets or channels. In contrast, innovating in business models implies a departure from how the company currently does business.

In this report, we explore the role of innovation on business models in financial services. We review innovation theory, concepts and frameworks to identify the innovation drivers in financial services. By integrating innovation and business model frameworks into one tool for analysis, we are able to discuss business models from business, innovation and technology perspectives. The business models that are analyzed in this report are selected on the basis of their potentially degree of impact on the financial services landscape.

Section 2 describes an overview of the current financial service industry landscape. It provides the context of the new economic environment in which innovative business models take place. Section 3 introduces innovation concepts and theories in the financial services industry domain. Section 4 discusses frameworks to analyze business models. Section 5 introduces our method to analyze the set of selected innovation-driven business models. In Section 6, we introduce the analysis of the selected set of innovation driven business models in the financial service industry. We end this document with conclusions.

2 Overview of the financial services landscape

The financial services industry has been traditionally a conservative industry that resists changes. Financial services industry used to have a stable structure with defined boundaries and clear business models. The stable set of players in the financial services made change linear and predictable in the past. In the last 20 years the financial services landscape changed significantly due to new market entrants and innovative business models [2].

The drivers of change in the financial industry includes advances in transaction and information technologies, geographic shifts in growth opportunities, regulatory changes, the fast evolution of client requirements and behavior. The transition to the changing landscape of financial services is a challenge for companies, requiring innovation as a source of profitable growth [2], [6].

This section presents an the overview of the current financial services landscape. We use a business model environment framework to analyze factors that affect business models in the financial services. We selected this framework because is an adaptation form management literature to analyze business models environments [7].

The business model environment framework identifies four main areas which are the key external forces (shown in Figure 2) that influence the business models in the financial industry. The market forces, industry forces, key trends and macroeconomic forces [7].



Figure 1: Business model environment diagram [7]

2.1 Market forces

We identify the market forces that are influencing the business models in financial services to perform a market analysis of the industry. The market forces that are shaping the financial services industry are depicted as follows: **Market issues:** One of the key issues driving and transforming the financial services market is the entrance of new competitors into the sector. The sectors that are more susceptible to new entrants are insurance, pensions, leasing and credit card companies. Another key market issue is that financial service companies have started to diversify into non-banking businesses [8].

Market segments: The market segments in the financial industry are categorized by their target clients (e.g : retail, corporate, institutional), service types (e.g : investment banking, retail banking, wealth management) and geographical location (e.g : Europe, Asia, Americas) [8].

The three dimensional representation of the market is used to recognize and maximize "synergies". Client driven linkages exist when financial institutions can supply services more efficiently to a client group in the same or other geographies. Service driven linkages exists when a financial institution already sells the same or similar service in other client or other geographic dimensions. Geographic linkages exists when a financial provider can supply services more efficiently in a particular location as a result of having an active relationship with that client in other geographic place [6].

Needs and demands: People without bank accounts and access to credits are underserved by financial services companies. New financial services by non-financial institutions are being developed to meet their demands [9].

Switching costs: Financial service customers have high switching costs to competitors. The customer lock-in is stronger when the financial service companies are based on a relationship rather than transactions. Customers find difficult to close bank accounts and refinance loans to competitors. Empirical evidence shows that first mover advantage is important in financial services, once a consumer learns to use a innovative product or service. Consumers do not want to switch to another provider due to the potential hassle of new learning [10], [8].

Revenue attractiveness: Financial services firms dominate the list of the world's biggest firms. Equipment leasing is recognized as the most attractive profit generating activities for financial services holdings [8], [11].

2.2 Industry forces

The identification of the industry forces behind the financial services sector help us to generate a competitive analysis of the financial services industry. The industry forces are composed by incumbent competitors, insurgent players, substitute products and services, value chain actors and stakeholders [7].

Incumbent competitors: Incumbent competitors are traditional companies in the financial services like retail banks, private banks, speciality finance and asset managers [12].

Insurgent players: Insurgents in the financial industry are the new entrants into the market. The insurgents in this industry are web-based financial services start-ups companies, telecommunication companies, retailers, car manufactures and industrial corporations [13], [8].

Substitute product and services: Informal loans by friends and family, and informal credit providers for the poorest households [14].

Stakeholders: Governments have a strong stake in the stability of financial services to sustain their economic activities. People and companies are important stakeholders, financial institutions depends on them to provide services [6].

Suppliers and other value chain actors: Examples for value chain actors in the financial industry are investment research companies, software development companies, IT infrastructure and support providers, and consultancy service providers [15].

2.3 Key trends

The recognition of the key trends in the financial services industry give us a foresight vision in technology, regulation, socio-economic, societal and regulatory aspects [7].

Technology trends: Technologies which could threat, enable, improve or evolve business models in the financial services are the internet and the telephone as distribution channels. A quarter of the world's population of 6.7 billion of people use internet and the mobile phone access is available to 90% of the world population [8], [16].

Social networks became a mass communication tool, which "made people's personal relationships more visible and quantifiable than ever". Twitter, was the fastest growing social network in 2008. Facebook, the world's larges social network, is the second most popular site after Google [17].

Regulatory trends: We can observe an increase of government intervention and a rising regulatory complexity in the financial sector [12], [18].

Societal and cultural trends : The major societal trend in the financial services is the eroded public trust on the financial industry. The consumer is more informed and collaborative, feeling the desire to contribute and being part of a community [19], [18].

Socioeconomic trends: The major socio-economic trends are given by ethical investment funds, green mortgages (for houses using renewable energy) and banking offerings for underserved communities. Companies committed to their corporate values and ethics, demonstrating good corporate citizenship which will be also necessary to recruit next generation employees [18], [16].

2.4 Macroeconomic forces

The identification of the macroeconomic forces help us to understand the overall conditions of the market [7].

Global market conditions: The global market conditions of the financial services market are a result by restricted credit availability and negative economic outlook. Emerging economies like Brazil, Russia, India and China (BRIC) are growing faster than developed economies. Global financial services providers are expanding their operations to the BRIC markets for the massive potential of consumers, workforce and high GDP [12], [2].

Capital markets: New entrants into the financial services industry are faced to the availability of lower venture capital and restricted access to credit [12], [7].

Commodities and other resources: The main commodities and resources needed for business models in financial services are information technologies (IT) and highly skilled human resources (HR). IT is more affordable thanks to server virtualization. HR cost depends on each region. In developed countries, IT and business people cost are higher than in emerging economies [8].

Economic infrastructure: Financial services are present in global markets, each specific region in which a company operates has a different economic infrastructure. Access to telecommunications services are different in the United States and the European Union from underdeveloped regions. For example, in Africa the access to internet is not common for the population. These unique conditions in different markets, like public infrastructure, education quality, public services and quality of life must be considered as factors that influence a business model [7].

3 Financial services innovation framework

This section describes innovation concepts, theory and classifications relevant for the financial services industry. The content is based on the book "Innovation and the future proof bank" by James Gardner [13]. We selected this source as a guide because offers a structured approach to innovation in financial services that is not present in other works (in Section 7, we review the related literature).

3.1 Innovation dimensions

Innovations in the financial services can be classified in two orthogonal dimensions: the newness dimension and the competitive dimension [13]. In the next subsections we explain each dimension and the kind of innovations that are part of it.

3.1.1 Newness dimension

The newness dimension is related to the degree of newness of the innovation, i.e, how the new creation is compared with previous innovation. In this dimension we can distinguish three kinds of innovations: breakthrough innovations, revolutionary innovations and incremental innovations.

Breakthrough innovations are concerned with the exploration of new technologies which have a high growth potential, but also imply higher risk. The risk is attributed to the lack of experience with the innovation that departs from the established offer of the company and its knowledge of proven business practices. These innovations may change directions of entire industries or create new ones due to the unpredictability of their scope, dimensions or economic effects [2], [20], [21].

The introduction of computing to financial services by Bank of America and Stanford University is a well-known breakthrough innovation. In 1950, banks began to struggle with high volumes of paper processing due to the introduction of credit cards. Later in 1955, a machine known as ERMA, Electronic Recording Method of Accounting, in conjunction with the Magnetic Ink Character Reading (MICR) technology enabled the cheque processing. This breakthrough innovation changed the way banks did business using machine processing instead of manual processing and by 1965 all banks in U.K and U.S.A were using systems like ERMA.

Revolutionary innovations are superior to what they replace, becoming the standard choice for a relevant market share. Revolutionary innovations do not create a new category in the market like breakthrough innovations do. Revolutionary innovations also have less entry barriers than breakthrough innovations, because they can be copied more easily.

ING Direct is an example of a revolutionary innovation. They were not first branchless internet banking, but offered a low cost service. ING Direct innovated with a no charge high interest saving account for low margin customers. **Incremental innovations** also known as continuous innovations are minor changes, exploiting existing technology. These kind of innovations are focused on cost or feature improvements in existing products, services, or processes. Incremental innovations are specific to an organization in the way of doing things which can be planned systematically [2].

An incremental innovation example is the mobile phone top-ups for Automated Teller Machines (ATMs). The mobile phone top-ups enable customers to add credit to pre-paid mobile phones from their bank account, entering their mobile phone number into the ATMs. This innovation uses what is already in place, the ATMs, and with minimal change in the functionality offers a new service.

3.1.2 Competitive dimension

The competitive dimension is related to relationship between the innovation in the firm and its competitive position. In this dimension, we distinguish two kinds of innovations: sustaining innovations and disruptive innovations. These innovations could sustain or disrupt the operations of institutions and markets, as we explain with the failure framework.

Sustaining innovations create additional value for a firm by enhancing the products or services that are already being offered. Sustaining innovations are those that resulted in an improved performance along the traditional value measure of the current market. Incumbents use sustaining innovations to differentiate among competitors to charge more or win more customers in the established market [22].

Sustaining innovations increase the capabilities of the current offer to make a more appealing value proposition to the more demanding customers which are willing to pay high prices to get such capabilities [22].

An example of sustaining innovation is internet banking. Internet banking is used by traditional companies to sustain their current business model [23].

Disruptive innovations usually starts as a poorly performing inferior product or service which does less than the current ones. Disruptive innovations are poor performers, because are measured against what historically matter in the mainstream market. Disruptive innovations are focused on convenience, simplicity, affordability or accessibility, creating new markets or transforming the current market [2], [22], [24], [2].

The disruption of the current market is made by targeting new or less demanding consumers. The new value proposition delivered by the new entrant is attractive to a small segment of the market, which is usually unattractive for the established firms [22].

A disruptive innovation in the financial industry is the Peer-to-Peer lending business model created by the U.K company Zopa. Zopa stands for "Zone of Possible Agreement" which refers to the price point reached when the borrow and lenders agree in the interest rate [13]. A deeper analysis of the Peer-to-Peer lending business model is provided in Section 6.1.1.



A : Overshoot value offered to high end customers of incumbents.

B : Undershoot value needed by the new entrant to attract low end customers of incumbent

C : New entrant is able to attract core customers of incumbent with a better value proposition

D : New entrant is able to attract all customers of incumbent with a better value proposition

Figure 2: Disruptive innovation mechanics in the financial industry [13]

Failure framework The disruptive innovation was identified by Clayton Christensen. Christensen introduced the failure framework to explain how disruptive innovation works. Figure 2 shows the disruptive innovation mechanics in financial services industry, which is an adaptation of Christensen's failure framework for the financial services industry made by Gardner [25], [22].

In Figure 2, "Arrow 1" represents the value proposition offered by the incumbent institution through sustaining innovation. Due to the pace of technological progress, incumbent's value proposition overshoots the value required by high net worth customers (wealthy customers with capital) to differentiate among competitors and charge higher prices. The overshot value is represented by "A" in Figure 2 [22]. "Arrow 2", shown in Figure 2, represents the new value proposition offered by the new entrant generated by disruptive innovation. The disruptive innovation of new entrant starts as a lower value proposition than incumbents. New entrant undershoots value to attract low net worth (low income) customers of incumbent (represented by "B" in Figure 2) [22].

Through sustaining innovation, the new entrant is able to offer a better value proposition than the incumbent and gain an established position in the low-end market. This behavior is shown in Figure 2 by the intersection point "C" of the dotted line that represents the low net worth customer's demand and the value proposition of new entrant [22].

After the disruptive innovation gains an established position in the new or low-end market, the new entrant starts to reach the demand of high net worth customers. We can observe this behavior in Figure 2, where the disruptor starts moving from point "C" to point "D". Then, the less performing innovation improves enough to satisfy the needs of more demanding customers, reaching point "D" in Figure 2 where the products and capabilities demanded by high net worth consumers are satisfied [2], [22].

A good example that is not related to the financial services industry, but useful to understand this failure framework is the video game hardware industry. Since the introduction of Atari, video games consoles evolved improving their power processing capacity. The PlayStation 3 is a sustaining innovation for Sony, because enhances a product that is already being offered (PlayStation 2). The traditional measure of the value is the graphics performance and the amount of polygons represented on the screen. Sony used this high processing power of PlayStation 3 to differentiate among the competition (Microsoft and Nintendo). Nintendo Wii is a disruptive innovation, changing the way people play video games by innovating the gaming controllers. If you compare the Nintendo Wii with the PlayStation 3 in processing power dimension, Nintendo Wii is a lower performer. Nintendo Wii changed the value in the video games industry. The disruption of the market is clear, Wii is leading the sales of the current generation of video game consoles. Nowadays, Sony and Microsoft are developing motion controllers to incorporate into their platforms the innovation introduced by Nintendo.

3.2 The innovation pentagram for the financial services industry

In this Subsection, we use the innovation dimensions explained in Subsection 3.1 to describe different innovation approaches that are used in the financial services industry.

The innovation pentagram shown in Figure 3 is a framework designed to visualize different innovation opportunities in the financial services industry. Each triangle of the innovation pentagram for the financial industry represents a field for innovation opportunity: products, markets, experiences, channels and business models [13].

The gray semi-circle areas in each triangle of the innovation pentagram in Figure 3 represents the probability for disruptive innovations and the remaining white area of the each triangle represents the probability for sustaining innovations from the competitive dimension.

The dotted lines within each innovation opportunity triangle represents the likely mix of newness innovations. The innovations dimensions are labeled by a letter inside a circle for each delimited dotted area. Incremental innovation is labeled by the letter "I", revolutionary innovation is labeled by the the letter "R" and breakthrough innovation is labeled the letter "B".

Next, we describe the innovation approaches identified in the innovation pentagram for the financial industry as presented in [13].

3.2.1 Product innovation

The focus on product innovation is the main force of innovation in the financial services industry. Financial services companies are constantly developing new products and marketing to their clients to sustain the current business mod-



Experiences

Figure 3: Innovation pentagram of the financial industry [13]

els. These new products do not deliver a sustainable competitive advantage because they can be easily imitated by the competition [26], [27].

In the innovation pentagram depicted in Figure 3, we can see that product innovation is mainly driven by incremental innovations. These incremental innovations are achieved by the introduction of modifications to an existing line of products [26].

3.2.2 Market innovation

A financial institution is not present in every market and customer segment. These unexplored markets are an innovation opportunity to expand the revenues of financial services companies [13].

Market innovations are used by incumbents to sustain their current strategy rather than disrupt the market. Innovating on markets is mostly incremental, because financial service companies find easier to adapt a current innovation to a new market rather than to create a new category for them. Since most financial service companies do not innovate specially for a market, breakthrough and revolutionary innovations are less likely to happen.

Financial service companies are used to buy foreign companies or create new ones to enter into a new market to sustain their business models. In 1960s multinational banks emerged to support U.S industrial companies that started to expand geographically. U.S banks like Citibank, Chase Manhattan, and JP Morgan followed their corporate clients abroad to service them and then entering into additional business segments. Incremental innovation has ben used to improve their products and services using information technologies [8].

3.2.3 Experience innovation

Experience innovation is about focusing on the improvement of the interaction with the customers. The enhancement of the customer experience requires the

adoption of a customer oriented focus by the company (known as outside-in, meaning that the company takes care about the customer needs to enhance their experiences) [28].

Currently, the most common perspective used by organizations is the insideout view, which only includes the internal vision to deliver a product or a service without the consideration of the customer needs. When the company considers the customer, they see the customer as a co-creator, because the inputs of the customer are key components of the customer experience and not just a one-way offer from the company to the consumers [29], [27], [30].

The experience innovation in the newness dimension is equally probable for breakthrough, revolutionary and incremental innovations, because the innovation on the customer experience could be equally applied using any kind of innovation.

In the competitive dimension, the probability for disruptive innovation on the experience innovation is medium because the financial industry is still organized in separated products instead of in an unified customer experience way. Disruptive innovations that are focused on the customer experience could gain market share taking away the customer loyalty [30].

An example of experience innovation is the pilot program in interactive tables that is being developed by Logica and Rabobank in The Netherlands. The project explores the new interactive Microsoft Surface tables technology, a breakthrough innovation to enhance the customer experience in financial services. Microsoft Surface is a "multi-touch computer that responds to natural hand gestures and real-world objects, helping people interact with digital content in a simple and intuitive way" [31]. The solution is aimed at enhancing the customer experience, allowing bank's employees to explain mortgages and loans in an easy and interactive way. The interactive table experience innovation in the competitive dimension is a sustaining innovation. The interactive table experience sustains the banking business, differentiating from competitors and wining more customers in the established retail banking market [32] [33].

3.2.4 Channel innovation

Channel innovation is about reaching customers in innovative ways. Incumbents and insurgents in the financial service sector are using different channels innovations to reach customers. In the competitive dimension, the occurrence of disruptive or sustaining innovations is equally probable. In the newness dimension, breakthrough innovations is more probable because of the exploration of new technologies.

Revolutionary and incremental innovations of the newness dimension are equally probable to happen in the newness dimension [13].

An example of channel innovation that uses internet as a payment channel is the payment service in the Dutch market provided by iDeal. This innovation let customers to do on-line payments in electronic commerce stores using their bank account. In the newness dimension iDeal is a revolutionary innovation, because is a superior method of payment for the Dutch market. In the competitive dimension iDeal is sustaining innovation for the traditional banks [34].

3.2.5 Business model innovation

Financial service institutions can also innovate in business models. "A business model describes the rationale of how an organization, creates, delivers and captures value" [7]. Business model innovation is the capacity to reframe an existing business model in new ways that create new value for the customers. Business model innovation can be a path to gain competitive advantage if the model is sufficiently differentiated and hard to replicate for incumbents and insurgents [35], [36].

In the newness dimension, business models innovations are most feasible to achieve through breakthrough innovations. Revolutionary and Incremental innovations offer less opportunities to business model innovation. Breakthrough innovations, due to the exploration of new technologies, brings the best opportunity for companies to reframe their business models in new ways. Companies could achieve breakthrough innovations, but they tend to commercialize new technologies through their current business models.

In the competitive dimension, the disruption of the market is more feasible through business model innovation. Business model innovation is a key factor in harnessing a disruptive innovation in order to transform an industry [37]. The main problem for companies to innovate their business model is that the capability to disrupt themselves is quite low. Academic research shows that existing assets and business models are the barriers to business model innovation. The profit margins from disruptive innovations start relatively below the current sources of incomes. This fact makes companies to keep focused on their more profitable resources, leaving behind the possibilities of self-disruption [38], [39], [25], [22].

Business model innovation is vitally important, but difficult to achieve. Companies need to embrace an attitude toward business model experimentation. Some experimental models will fail, but provide information to new approaches. Organizational aspects will need to find ways to embrace a new business model, maintaining the effectiveness of the current business model until the new one is ready to fully take over [39].

An example of business model innovation is Peer-to-Peer lending created by Zopa (analyzed in Section 6.1.1).

4 Business model frameworks

In this section we depict business model frameworks, since there is a relationship between innovations (from Section 3) and business models. Financial services companies are using innovations as a driver to support or change their business model. Innovations are used to support the current business model of a company through the enhancement of their products, reaching new markets, revamping customer experiences or improving their channels. Innovating in business models implies a departure from how the company currently do business, which explain why this kind of innovation is pushed by new entrants into the financial services market and not by established companies.

Innovations without a business model will fail to deliver value to their customers. "The essence of a business model is in defining the manner by which the enterprise delivers value to customers, entices customers to pay for value, and converts those payments to profit" [36]. Alexander Osterwalder did an extensive literature review about business models definitions and concepts in his Ph.D thesis. He define a business model as : "A business model describes the rationale of how an organization, creates, delivers and captures value" [7], [40].

In this section, we describe two selected frameworks to provide complementary visions. The first selected framework, the business model canvas by Alexander Osterwalder, is a visual representation to describe a business in an easy to understand manner. We selected this method because is based on an extensive research about business models and is currently used in organizations such as IBM, Ericsson, Delloite and many more [39], [40], [7].

Since financial services are being under a major transformation due to ebusiness, we need a specially designed a framework focused on the impact of e-business elements on business models [41]. The second selected framework is the business model for e-business by Paul Grefen. We selected this method because provides specific ingredients for a business model like business drivers, chains, directions and structures. This components are important to consider with financial service companies, because they do businesses as financial intermediaries. For example, using business chains we can analyze the disintermediation and reintermediation of financial services.

The two frameworks are complementary. With the business model canvas we can represent any kind of business, but we can not address in detail e-business elements. The business model canvas, complements the e-business vision because we can provide a better overview of the business model, thanks to the visual representation of the four business areas. The business for ebusiness complements the business model canvas, including specific e-business components that are needed to analyze innovative business models in the financial services industry.

4.1 The business model canvas

The business model canvas describes a business model through nine basic building blocks that show the logic of how a company intends to make money. The nine blocks cover the four main areas of a business: customers, offer, infrastructure and financial liability. The four areas are influenced by the Balanced Scorecard (BSC) of Kaplan and Norton, and more general management literature. The BSC influences the four areas as follows: the customer area is influenced by the customer perspective, the offer area is influenced by the innovation and learning perspective, the infrastructure area is influenced by the internal business perspective and the financial area is influenced by the financial perspective. The customer area includes three building blocks: customer relationships, customer segments and channels. The offer area includes the value propositions building block. At last, the financial liability includes two building blocks: cost structure and revenue streams [40], [7]. The visual representation of the business model canvas is shown in Figure 4.



Figure 4: Business model canvas [7]

As presented in [7], the building blocks are:

- Customer segments: "defines different groups of people or organizations an enterprise aims to reach and serve".
- **Value propositions:** "describes the bundle of products or services that create value for a specific customer segment".
- **Channels:** "describes how a company communicates with and reaches its customer segments to deliver a value proposition".
- **Customer relationships:** "describes the types of relationships a company establishes with specific customer segments".
- Revenue streams: "represents the cash a company generates from each customer segment".

- Key resources: "describes the most important assets required to make a business model work".
- Key activities: "describes the most important things a company must do to make its business model work".
- Key partnerships: "describe the network of suppliers and partners that make the business model work".
- Cost structure: "describes all cost incurred to operate a business model".

4.2 Business model for e-business

The business model for e-business framework is composed by the e-business classification space and the business (B) aspect of the BOAT framework. In Figure 5, the ingredients are shown to analyze a business model in the e-business. The vertical lines represent the classification dimension that are useful to categorize business models. The classification dimensions are parties, objects and time scopes. The horizontal line represent the B aspect and its components are business drivers, business chains, business directions and business structures.



Figure 5: Business model for e-business

In the following sections we describe the business model fore-business components : the classification space and the B aspect of the BOAT framework.

4.2.1 E-business classification space

The classification space is a tool to classify business models in e-business in three orthogonal dimensions : parties, objects and time scopes. The dimensions are described as follows:

- Parties: Defines the parties that perform the e-business activities. The main party combinations that we can be distinguished are Business to Business (B2B) e-business, Business to Consumer (B2C) e-business and Consumer to Consumer (C2C) e-business.
- Objects: Defines the type of objects that are primary manipulated by e-business activities. The objects that we can distinguish are physical goods, digital goods, services, financial goods and hybrid objects (any combination of the above).
- **Time scopes:** Defines the duration of the collaboration between the parties involved on the e-business activities. The collaboration could be static, semi-dynamic, dynamic and ultra-dynamic.

4.2.2 B aspect of the BOAT framework

The BOAT framework is a method to analyze and build e-business cases using four aspects: business (B), organization (O), architecture (A) and technology (T). In this report we only use the B aspect of BOAT, because is the related aspect with business models. The O, A and T aspects are not discussed on this report because the business model for e-business is a subset of e-business cases.

The topics in the B aspect could be the leverage of efficiency levels, access to new markets, reorientation of interaction with customers, etcetera. This aspect is focused in the why question, how things are done is not of interest in this aspect [42]. The B aspect components are described as follows:

- Business drivers: Contains the essential for contacting business parties, engaging into business with them and retaining them as a parters using information and communication technology (ICT). Through ICT we can increase reach and richness.
- Business chains: Describes the restructuring collaborations in a business chain like disintermediation, reintermediation, reconstruction and deconstruction, and integrated bricks and clicks.
- Business directions: Depict the new business directions that can be distinguished in the e-business domain like true on-time and on-line capability, enriched Customer Relationship Management (CRM) and timecompressed e-business.
- **Business structures:** Covers the new business structures of collaboration with consumers and between business partners. The traditional approach of supply chain is changed to demand chains, highly supply chains and dynamic service outsourcing.

4.3 Summary of this section

In this section, we discussed two complementary business models frameworks. With the business model canvas framework, we can represent any kind of business model with a visual overview in an easy to understand manner. The business model for e-business complements this representation, including the specific e-business components. These components are needed to analyze insurgents with innovative business models, which are changing the business chains in the financial services industry. In Section 5, we use these two complementary frameworks to present our method to analyze innovation-driven business models in the financial services industry.

5 A method to analyze innovation-driven business models in the financial services industry

In this section, we describe an analysis method for business models driven by innovation. With our method, we can analyze a business model from the innovation, technology and business perspectives. The method is based on the combination of three frameworks.

The innovation perspective is needed to identify the innovation driver that changes or create business models. The innovation driver can be identified using the innovation pentagram for the financial services from Section 3.2, where we categorize the innovation by different kinds of innovation opportunities and by the competitive and newness dimensions from Section 3.1.

With the business perspective, we provide a complete overview of a business model. This perspective is described using the business model canvas from Section 4.1 which includes nine building blocks to provide a broad view in a simple manner.

In the financial services industry electronic business is being used by incumbents to support their business and by insurgents to disrupt the market. The technological perspective describe the electronic business elements needed using the business model for e-business framework from Section 4.2.

Our method contains three steps for the analysis of innovation drivenbusiness models. The steps are as follows:

- 1. **Apply the business model canvas:** Describe the business model using the canvas from Section 4.1. Identify the nine building blocks and produce a visual representation as shown in Figure 4.
- 2. **Apply the business model for e-business:** Identify the components of the business model in e-business from section 4.2. Fill the components shown in Table 1 (listing the relevant e-business components).

Business model for e-business		
Parties		
Objects		
Time scope		
Business drivers		
Business directions		
Business chains		
Business structures		

Table 1: Business model for e-business

3. **Identify the Innovation driver:** Categorize the innovation opportunity using the innovation pentagram for the financial industry and categorize the innovation in the competitive and newness dimensions from Section 3. Fill the innovation driver summary, shown in Table 2 (listing the rele-

vant innovation concepts).

Innovation driver
Innovation opportunity
Competitive innovation dimension
Newness innovation dimension

Table 2: Innovation driver summary

The method presented is used in Section 6 to analyze the emerging innovationdriven business models in the financial services industry.

6 Emerging innovation-driven business models in the financial industry

In section 5, we described a method to analyze innovation-driven business models in the financial industry. In this section, we present a set of business models driven by innovation and analyze them using our method. The selection of the business models is made on the basis on their impact on the current financial industry [43], [13].

6.1 Peer-to-Peer lending

Business model canvas: The concept of borrowing money between people is not new. Even in the year 2000, Tapscott predicted the arrival of new internet based intermediaries referred by him as "investormediaries". These "investormediaries" would be capable to manage lending at very low cost, inviting depositors to lend money to borrowers [44].

In deed, as predicted new electronic intermediaries emerged. Peer-to-Peer (P2P) lending, also known as social lending, is an emerging alternative to banks and personal loans that allows individuals to lend/borrow money to each other directly [45], [44].

The value proposition of P2P lending to borrowers is the opportunity to obtain loans at lower interest rates and costs. In P2P lending we can distinguish between two consumer segments: the borrowers and the lenders. A borrower is viewed as an investor. The investor's money is divided across different lenders to distribute the risk. For lenders, the value proposition of P2P lending relies in the investment opportunity with higher rates of return on investment than traditional financial institutions [45]. The P2P lending business model is visualized in Figure 6.

Key Partners	Key Activities	Value Proposition		Customer Relationships	\mathcal{Q}	Customer Segments	
Credit Scoring companies Platform Maintenance and development Profitable Investment opportunities		community		Wealthy investors			
Loan processing banks	Resources P2P Lending Platform	Eas acc t cre	sier ess o dit	^{channels} P2P Websi	te	Borrov withc traditio acce to cre	vers out onal ss edit
Cost Structure IT infrastructure Software Partners Developments fees			Revenue Streams Percer Percen	ntage or fix tage fees f	ked fee or rec	es for loar eived pay	ns ments

Figure 6: P2P lending business model canvas

The P2P lending business model uses a website to reach their customers. The key asset of the P2P lending business model is the lending platform that mediates between borrowers and lenders [46].

The revenue streams of P2P lending companies are based on fixed fees or a percentage. The percentage fee for borrowers is between 0.75 % and 3.5 % of their loan amounts. For lenders, the percentage fee is between 0.5 % and 1% of the payments received [46].

The cost structure fits concentrated in the maintenance and development of the platforms and the payment for the key partners services. These key partners companies are credit reporting institutions and loan processing banks.

Business model for e-business: P2P lending is a consumer-to-consumer (C2C) business model. The parties are consumers who exchange financial objects (money). The time scope is static, because P2P lending partners are the same for every user. The business drivers of this business model are the increased reach that let users to borrow/lend financial objects and the rich interactions that are developed between borrowers and lenders [47]. The business direction of this business model relies in the true on time and on line capability of the platform to exchange the financial objects between people that is used to disintermediate and reintermediate the traditional credit offered by incumbent financial companies. Business structures are traditional supply chains. A summary of the P2P lending business model for e-business is shown in Table 3.

Business model for e-business			
Parties	C2C		
Objects	Financial		
Time scope	Static		
Business drivers	Increased reach and richness		
Business directions	True On-time and On-line		
Business chains	Disintermediation and reintermediation		
Business structures	Supply chain		

Table 3: P2P lending business model for e-business

Innovation driver: P2P lending is a disruptive, breakthrough, business model innovation. P2P lending is disruptive because is a lower performer in the financial industry when is compared with traditional financial services institutions. In the newness competitive dimension, P2P lending is a breakthrough innovation because this innovation creates a new category in the lending industry. A summary of the innovation driver is shown in Table 4.

Innovation driver		
Innovation opportunity	Business model	
Competitive innovation dimension	Disruptive	
Newness innovation dimension	Breakthrough	

Table 4: Innovation driver of the P2P lending business model

In the following subsections we give two examples of the P2P lending business model : Zopa and Lending Club.

6.1.1 Example 1: Zopa

The name Zopa comes from the negotiation theory concept "Zone of Possible Agreement". The company discovered the need of an undeserved market. The "freeformers" are self-employed, project-based or freelance workers that are not in a full time employment with irregular incomes and lifestyles [48].

Zopa adds value to consumers by eliminating the need to work with traditional financial institutions for obtaining credit. It eliminates the requirement for the many face-to-face interactions and manual processes that have traditionally been part of borrowing. The company also raises flexibility and transparency for customers well above the industry standard, as customers can borrow smaller amounts over shorter periods and are not charged additional fees if they repay early [48].

Zopa tried to lock down its key asset based on a proprietary marketplacematching platform, but it recognizes that the business model concept of P2P lending can not be exclusive [48].

6.1.2 Example 2: Lending Club

Lending Club offer proposition is based on being cheaper and faster than traditional consumer credit. For example, Lending Club's rate for the best credit risks is 7.88%, whereas the bank rate for personal loans, on average is over 13%. A credit-worthy borrower gets the money faster and for 5% less. Lending Club starts with traditional credit scoring and adds a proprietary assessment of customer reputations within their social networks [47]. Lending Club is unique in that it makes nearly all that information public (aside from data that could lead to privacy concerns), tracking and publishing the history of every loan. Lending Club posted to its website the formula it uses to measure default risk and determine the interest rates its borrowers had to pay. Most banks keep in secret the risk assessment algorithm as a competitive advantage. Lending Club open-sourced their algorithm, asking readers to submit their own tweaks and improvements.

After receiving a slew of suggestions, the site's engineers decided to modify the equation, assigning less weight to debt-to-income ratio, for instance. Other Lending Club lenders downloaded the equation and came up with their own proprietary improvements, devising a better formula so they could cherry-pick borrowers who were wrongly categorized as risky and charge them higher interest rates without worrying about defaults. All this innovation benefited not just individual lenders but the entire ecosystem. Lending Club's default rate is a staggeringly low 2.7 % (versus nearly 5.5 % for prime credit cards) [49].

6.2 Social investing portfolio

Business model canvas: The value proposition of social investing portfolio business model is to offer transparent records for people who want to invest by following and replicating the trades made by successful investors. The investment records are transparent, because the followers can always see the moves

of a successful investor leader. A visual representation of the social investing portfolio business model is shown in Figure 7.



Figure 7: Social investing portfolio business model canvas

The customer segments are investors and followers. Investors who share investment strategies with an outstanding performance in stock trading and followers who are seeking for better investment opportunities. The customer relationship is made by the interaction of the investment community (investment leaders and followers) through the social investment website channel. The customer relationship is made by the interaction of the investment community between the investment leaders and followers through the channel represented by the social investment website.

The key resource is the social investment platform in which key activities like maintenance and development happen. The key partner is an on-line brokerage service, which provides the back-end for automated transactions that are triggered by investor's followers. The revenue stream comes from the percentage fees charged to the investment followers. The cost structure is driven by the platform maintenance and development, the percentage fees that are paid to the investor leaders for sharing their stock trading moves and the fees paid for the on-line brokerage partners that execute the transactions.

Business model for e-business: The social investing portfolio is a C2C business model. The parties involved are consumers who trade financial objects (stocks). The time scope is static, because the relationship with the on-line stock brokers is long-lasting.

The business drivers of this business model are increased reach delivered by the web-channel and the richness on investment opportunities and information available. The social investment portfolio business model changes the business chains disintermediating and reintermediating the access to investment experts that was previously available only for wealthy investors through private banking. The business directions are true on-time and on-line capability, because the follower user can always look and stop following a leader investor at any time. Time-compressed is also a business direction, since the social investing platform automatically triggers the transactions made by a investor for a follower. The business structure of this business model is the standard supply chain provided by the on-line brokerage partners to accomplish the automated transactions in the social investment platform. A summary of the social investing portfolio business model for e-business is shown in the Table 5.

Business model for e-business			
Parties	C2C		
Objects	Financial		
Time scope Static			
Business drivers	Increased reach and richness		
Business directions	True on-time and on-line		
	Time-compressed e-business		
Business chains	Disintermediation and reintermediation		
Business structures	Supply chain		

Table 5: Social investing portfolio business model for e-business

Innovation driver: The social investing portfolio business model is driven by business model innovation opportunity. The innovation is breakthrough, since it establishes a new category in the market. The innovation is disruptive as it introduces transparent investment which was not offered by traditional investment companies like mutual funds. The disruptive innovation started as a low performer, for example KaChing a social investment portfolio company started as a Facebook game about fantasy stock market. A summary of the innovation driver of the social investing business model is shown in the Table 6.

Innovation driver			
Innovation opportunity	Business model		
Competitive innovation dimension	Disruptive		
Newness innovation dimension	Breakthrough		

Table 6: Innovation driver of the social investing business model

Next, we give two examples of the social investment business model : Covestor and KaChing.

6.2.1 Example 1: Covestor

Covestor is an investment site that recruits traders with strong track records, good returns and a defined strategy to become "model managers". Covestor allows users to follow these "model managers" and replicate their market moves of traders with strong tracks records [50].

Covestor brings a service that was perviously available through firms only to wealthy individuals as separately managed accounts (SMAs). SMAs are individual investment accounts overseen by professional money managers. Money managers of SMAs usually handles clients who are seeking a particular investment style or strategy with investments between \$100.000 to \$250.000 U.S dollars per account who pays from 1% to 3% of account assets [51].

Covestor formed Covestor Investment Management (CVIM) a SEC registered investment advisor (RIA) that uses a separate brokerage account from TD Ameritrade Holding Corp. or Interactive Brokers Group Inc. to perform the actual trades that are linked to a "model manager" [51]. Covestor charges between 0.5 % to 1.5 % of traded funds [50] and the minim investments starts at \$10.000 U.S dollars [51].

6.2.2 Example 2: KaChing

KaChing is an investment site that lets professional investors and amateurs post their stock trades [52]. Each investor has its own investment IQ and strategy that are used to decide if they qualify as an outstanding investor which are known as Genius with the goal to earn money as stock trading advisors. To qualify as a Genius, an stock investment advisor, an IQ of at least 140 is required with an investment track record of at least twelve months in which no single position in the portfolio represents more than the 25% of the market value [53]. KaChing let users to follow or mirror the investments made by the Geniuses. Kaching ask for a the minimum amount of \$3.000 dollars to perform the mirror transactions of a Genius. The mirroring of the genius stocks are processed by their key partner Interactive Brokers [54] in which the automated trades are made. KaChing charges customers a single management fee between 0.25% to 3% set by each investor, keeping a quarter of the fee and the rest for the investor [54].

6.3 On-line personal finance

Business model canvas: The value proposition of the on-line personal finance (OPF) business model is the enhancement of customer experience that provides an easier way to manage expenses to save money. A visual representation of the OPF business model is shown in Figure 8.

The customer segments are the web savvy customers that want to save money. OPF uses a website and a smartphone application as channels to reach the customers. The cost structure is given by the platform development maintenance since they do not have physical facilities. The customer relationship is achieved through communities, where people access to aggregated data to compare their personal expending.

The key partners are banks, credit card companies and stock trading companies that use the platform to advertise their products. The OPF revenue streams comes from the payments on the recommendations that are made to the users by referred companies or by providing the service to banking and credit card companies.

The key resources of the OPF are the platform and the consumer data. The key activities of the OPF business model are the analysis of user data to compute the spending trends of the community and advise the users about the



Figure 8: OPF business model canvas

lowest interest rates of financial service providers.

Business model for e-business: The OPF is a B2C business model when the users access directly the service and a B2B business model when the company sells OPF as a service for banks. A summary of the OPF business model for e-business is shown in Table 7.

Business model for e-business			
Parties	B2C, B2B		
Objects	Financial		
Time scope Static			
Business drivers Increased reach and richness			
Business directions True on-time and on-line			
Multi channel business design			
Business chains Disintermediation and reintermediation			
Business structures	Supply chain		

Table 7: OPF business model for e-business

The parties involved in OPF are businesses and consumers who trade financial objects (money). The time scope of this business model is static, because the relationship of the partners involved in the collaboration is long-lasting. The business drivers of the OPF business model are increased reach through the web and smartphone channels that provide an increased richness. The increased richness is achieved thanks to the high level of interactivity that is provided by the manipulation of information and interaction with other users. The OPF disintermediate user's access to regular electronic banking websites because they gather and display the information from traditional banks, creating a new link in the chain between the user and their financial data [13]. The business directions of this business model are driven by the true on-time and on-line capability that gathers information from the traditional financial services accounts of the user and by the multi-channel business design that allows users to interact with the service through a smartphone application that is being synchronized with the OPF website.

Innovation driver: The OPF business model is driven by channel innovation. The competitive dimension, is disruptive innovation because is a lower performer compared to the features provided by financial desktop software and transactional electronic banking. OPF companies altered the finance software management industry forcing Microsoft to discontinue the production of the Money Plus desktop software [55]. OPF influenced banks, forcing them to integrate similar features into their electronic banking platforms to help customers to visualize their expending habits. OPF is a breakthrough innovation because created a new category previously dominated by desktop software.

A summary of the innovation driver of the OPF business model is shown in Table 8.

Innovation driver			
Innovation opportunity	Channel		
Competitive innovation dimension	Disruptive		
Newness innovation dimension	Breakthrough		

Table 8: Innovation driver of the OPF business model

In the following subsections we give two examples of the OPF business model. Mint.com is the leading OPF company and Wesabe is a well known follower.

6.3.1 Example 1: Mint.com

Mint.com, an OPF start-up is recognized by the World Economic Forum as a pioneer. The start-up provides a free on-line money management service designed to help users to save money more easily. The service was launched officially in September 2007. In October 2009, mint.com had more than 1.5 million users. Users register anonymously, with only a valid email address and a postal code. The users need to provide Mint.com with the login details to all their bank accounts. By connecting to more than 7,500 U.S financial institutions, Mint.com applies technology to unscramble the transaction descriptions found on credit card, bank and brokerage statements. Mint.com processes data daily into neat graphs of cash flow and expenditure. Purchases are colorfully categorized to show how much a user spends in the pub, on parking, on rent or in restaurants. Users get a dashboard with their investment performance and fees that are clearly displayed [56].

The OPF Web site provides recommendations to their users to save money using cheaper credit cards, based on their own spending patterns. Mint.com makes its revenue from these referrals, obtaining fees from banks, brokers and other financial institutions. This service also alerts users when their bank balances are getting low or have any overdue bills. Mint.com also alerts about any potential suspicious activity on the customer accounts [56].

Another interesting Mint's feature is the "ways to save". This feature makes behavioral targeted advertising to customers based on credit score, purchasing history and other financial metrics. "Ways to save" is used by 20% of mint.com's customers.

6.3.2 Example 2: Wesabe.com

Wesabe is an OPF company that lets users visualize and track their spending automatically from over six thousands bank accounts and credit cards or manually by importing financial file types such OFX and QIF [57].

The service includes innovative comparison tools for spending habits, including a community platform which let users share tips on how to save money. The community interaction has three key features: Tips, Goals and Groups. The Tips feature can be used by users to get information about similar retailers and satisfaction ratings from their transaction information. The Goals feature can be used by users to track spendings in a specific tag where you can see and interact with other community members to share tips and experiences. The Groups feature is the common on-line forum where users can interact on s different topics [57].

Wesabe makes its revenue from offering a OPF solution for banks and credit unions called Springboard. For example, the platinum plan costs \$1.799 dollars per month for 120.000 users [58].

6.4 Electronic payment platform

Business model canvas: The value proposition of the electronic payment platform (EPP) is to provide a flexible payment infrastructure that can be accessed by any software developer.

The customer segments are startup companies, e-commerce websites and independent developers. EPP deliver flexibility, allowing their customers segments to incorporate innovative payment schemes into their business models. The customer relationship is engaged on the EPP's website channel where the developers can learn about the service and interact with other members of the community.

The key resource is the payment platform that is open to anyone who wants to implement an e-commerce solution. The key activities are the maintenance, development and documentation of the key resource represented by the electronic payment platform. The key partners are the developers and e-commerce partners that implement solutions, delivering the brand awareness of the payment providers across the internet.

The revenue streams come from transaction fees over the payments. The cost structure is given by the transaction fees from credit cards companies and banks and the platform maintenance, development and documentation.

A visual representation of the EPP business model is shown in Figure 9.

Business model for e-business: The electronic payment platform is a B2B business model. The parties involved are businesses who use the platform to



Figure 9: EPP business model canvas

trade financial objects. The time scope is static because the service offered is the same for all the business partners.

The business drivers are represented by an increased richness on the functionality and an increased reach through the internet channel. The business directions are true on-line and on-time access to the payment platform. The changing business chain is driven by the disintermediation and re-intermediation of the payment services with a powerful platform. The business structure is determined by the traditional supply chain.

A summary of the EPP business model for e-business is shown in Table 9.

Business model for e-business			
Parties	B2B		
Objects	Financial		
Time scope	Static		
Business drivers	Increased reach and richness		
Business directions	True on-line and on-time		
Business chains	Disintermediation and re-intermediation		
Business structures	Supply chain		

Table 9: EPP business model for e-business

Innovation driver: The innovation opportunity of the EPP business model is channel innovation. The innovation is disruptive, because a specific payment functionality is offered targeting start-ups developers. The innovation is breakthrough, because the solution is a complete new category on the market.

A summary of the innovation driver of EPP business model is shown in Table 10.

In the following subsections we give two examples of the EPP business

Innovation driver		
Innovation opportunity	Channel	
Competitive innovation dimension	Disruptive	
Newness innovation dimension	Breakthrough	

Table 10: Innovation driver of the EPP business model

model: Amazon Flexible Payments Service and PayPal Adaptive Payments.

6.4.1 Example 1 : Amazon Flexible Payments Service (FPS)

Amazon FPS is a payment platform designed for clients who want to add a payment solution of their software. Amazon offers FPS Quick Start, which are sets of functionality to integrate payment transaction schemes into customer's websites. The Amazon FPS Quick Start are : Basic, Advanced, Marketplace and Aggregated [59].

Amazon FPS Basic Quick Start enables easy integration to websites that need to support one time payments through a payment button. Amazon FPS Advanced Quick Start provides input for a periodic or delayed payment required for subscriptions services like digital music or on-line storage. Amazon FPS Marketplace Quick start is designed for intermediaries that want to build their own marketplace, making it easier for them to have a share in a transaction made between a buyer and a third party seller. Amazon FPS Aggregated Payments Quick Start enables the developer to consolidate multiple transactions into a larger transaction to reduce payments cost, enabling end users to create prepaid balances that are used to make multiple smaller purchases or the developer can track multiple individual transactions and charge the aggregated amount later [59].

Amazon fees are per-transaction depending on the payment method used and the transaction amount. For example, the transaction fee for payments under \$10 U.S dollars that use credit cards as a payment method is 2.9 % of the transaction amount plus \$0.30 U.S dollars [59].

6.4.2 Example 2 : PayPal Adaptive Payments

Paypal is the most successful internet payment scheme [60]. PayPal is extending their coverage as a platform through their Adaptive Payments. PayPal Adaptive Payments is designed to handle the payments between a sender and one or more receivers of the payments. PayPal defines three kind of Adaptive Payments : simple, parallel and chained payments [61].

"Simple payments" allows a sender to send a payment to the receiver, where the money of the sender is transferred from the customer's PayPal account to the receiver's PayPal account. An example of the usage of this payment method is a customized payment button on the receiver's website [61].

"Parallel payments" allows to a user of a website to send a single payment to multiple receivers. This payment method is useful for a website that integrates several merchants. In this case, the payment of an user will be deducted and transferred to the receiver's PayPal accounts. "Chained payments" allows a sender to send a single payment to a single receiver who may keep part of the payment and send to secondary receivers the remaining amount. "Chained payments" are useful for an online travel agency which handles airline bookings, hotel reservations and car rentals. The sender sees the travel agency as the primary receiver that allocates a commission for the services. PayPal deduce the commission from the payments made by the user and deposits the remainder in the secondary receivers accounts [61].

6.5 Industrial equipment with telematics services

Business model canvas: The equipment with telematics services is not a financial service. However, since leasing companies finance equipments to their customers, this business model is relevant for them. The industrial equipment with telematics services (IETS) business model is focused on selling equipment that includes services based on telematics as a differentiator among the competition. Telematics is an asset management technology that helps industrial equipment customers like mining and rental companies to manage, monitor and maintain their physical resources. Nowadays, most major brands in the heavy construction equipment business are including telematics in their equipments [62]. A visual representation of the IETS business model is shown in Figure 10.

Key Partners	Key Activities	Value Proposition		Customer Relationships	\heartsuit	Customer Segments	A
Telematics provider	Platform maintenance and development	Maximize equipments investment with a telematics value added service		Service website Equipment dealers		Equip custor	ment ners
	Key Resources			Channels			
	Service platform			Equipm deale	nent rs	Equip ren compa	ment tal anies
Cost Structure Platform maintenance and development		Revenue Streams Equip subse	oment sale cription to t	s with the ser	a free or p vice	baid	

Figure 10: IETS business model canvas

The value proposition of the IETS business model is to maximize the return on investment (ROI) of industrial equipment through the value added service of telematics.

The key partner for the equipment provider on the IETS business model is the telematics partner who provides the asset management technology. The service is offered with the machine sales through the dealership channels which are targeted to equipment customers and rental companies that are attracted to the value proposition [62].

The key activities of the business model are related with the maintenance and development of the key resource, the service platform. The cost structure of this business model are the partner fees and the platform maintenance and development. The revenue streams are driven by the sales of equipments bundled with the service or by optional payments for the service.

Business model for e-business: The parties involved are businesses (B2B) who trade hybrid objects composed by physical equipments with value added services. The time scope is static, because the relationship between the partners involved in the business models is long-lasting.

The business drivers of this model are the increased richness offered by the information delivered to the customer and the increased reach of the service across the world. The business directions of the business model are based on the true on-time and on-line access to the service, where the customer can access any time the information system through a website. Integrated brick and clicks is another business direction of this business model, since the traditional equipment sales integrates an electronic service to monitor them. The changing business chains is driven by reintermediation adding a link chain to the current equipment selling business. The business structure of this model is the traditional supply chain that is used to offer the equipment bundled with services. A summary of the IETS business model for e-business is shown in Table 11.

	Business model for e-business				
Parties		B2B			
	Objects	Hybrid (physical and services)			
Time scope		Static			
Business drivers		Increased reach and richness			
Business directions		True on-time and on-line			
		Integrated brick and clicks			
Business chains		Reintermediation			
	Business structures	Supply chain			

Table 11: The IETS business model for e-business

Innovation driver: The IETS business model is driven by experience innovation, since the focus is on helping the customers to maximize their equipment ROI. The experience on the way the customer interact with the maintenance of the equipment is truly enhanced thanks by the usage of breakthroughs innovations (newness dimension) on telematics that let users know more about their assets. This business model is a sustaining innovation, because supports equipment manufacturers current business model based on selling equipment with value added services. A summary of the innovation driver of the IETS business model is shown in Table 12.

Next, we describe two examples for the IETS business model: Komatsu's komtrax and Volvo's CareTrack.

Innovation driver		
Innovation opportunity	Experience	
Competitive innovation dimension	Sustaining	
Newness innovation dimension	Breakthrough	

Table 12: Innovation driver of the IETS business model

6.5.1 Example 1: Komatsu's komtrax

Komatsu is an innovative company that is using services to add value to the equipment sales. Komatsu developed the Komtrax service that track the location of its heavy machinery anywhere in the world. The service also records how much fuel every vehicle consumes and the amount of strain on the most heavily used weight-bearing parts [63].

The original purpose of Komtrax was to monitor leased equipment to prevent theft. Komtrax lets Komatsu to know how much wear and tear its machinery is getting and when it should dispatch staff to perform maintenance to customers. The record is useful to calculate the resale value of vehicles that get traded in for new purchases [63].

Komtrax's biggest benefit is the real time snapshot it provides of construction activities on every country where Komatsu does business. The information retrieved from Komtrax gives Komatzu the ability to react to market shifts with real time information. Knowing the usage information they can forecast the demand and pass this information to their parts suppliers [63].

The Komtrax service let customers access to a Web application for machinery information (location, fuel consumption, maintenance alerts, daily and monthly status reports). Komtrax is being used at this moment in 144,000 machines. Komtrax partly explains why Komtasu has avoided losses despite the freezing in building and mining projects across the world [63]. The service supplied to the customers is free of charge for those who purchased the equipments with Komtrax [64].

6.5.2 Example 2: Volvo's CareTrack

CareTrack is a system to monitor equipment which provides mapping and tracking, operation reports and service management [65].

The mapping and tracking allows customers to track their machines in real-time with valuable information like fuel levels and usage. The reporting features helps customers to know about fuel consumption and usage time of the tools to know the real work made by the equipments. The service management helps the customer to plan the maintenance and look up for service history [65].

6.6 Money service provider for unbanked people

Business model canvas: The market segment of this business model is the people without access to banking services. "Unbanked people" is a potentially rich source of revenue for financial service providers [13]. The Consultative Group to Assist the Poor (CGP), housed at the World Bank, found that more

than a billion people worldwide had a phone but no bank account [66]. A study conducted by the "Group Speciale Mobile Association" shows that the estimated unbanked people that could be signed by mobile phone financial services by 2012 is up to 360 million [66]. A visual representation of the money service provider for unbanked people business model is shown in Figure 11.



Figure 11: Money service provider for unbanked people business model canvas

The value proposition is based on lower transaction fees than the current providers to transfer money or do payments for unbanked people with the ease and convenience of using a basic cellular phone. The channels used are the cellphones and physical points of sale (POS) to enable customers to make deposits and retrieve money. The customer relationship between the company and the customer is developed through the physical POS.

The key activities involved are Physical POS services and infrastructure management. The key partners for this business model are cellphone companies, cellphone manufacturers or network operators with traditional banking institutions like a bank or a debit card provider to store money and perform the financial transactions. The cost structure of this model is based on the platform maintenance, partner fees and the POS costs. The revenue streams are the transaction fees that are charged to the customers.

Business model for e-business: The parties involved in this business model are business and consumers who use financial objects to do payments and money transfers. The partners in this business model are static, since they have a long-term relationship to offer the service. The business driver is an increased reach thanks to the mobile phone devices that give a true on-line and on-time financial service access to unbanked people. Companies from the cellphone industry are using their resources to re-intermediate the banking chain [13]. A summary of the money service provider for unbanked people business model for e-business is shown in Table 13.

Business model for e-business		
Parties	B2C	
Objects	Financial	
Time scope	Static	
Business drivers	Increased reach	
Business directions	True on-line and on-time	
Business chains	Reintermediation	
Business structures	Supply chain	

Table 13: Money service provider for unbanked people business model for ebusiness

Innovation driver: The money service provider for unbanked people business model is a market innovation. The innovation is disruptive because is a low performer compared with the current services offered by banks. After gaining market share this innovation could be improved using sustaining innovation to please the high-end segments of the market. It is revolutionary, because brings a superior designed service for people with basic cellphones, becoming a standard choice for a relevant market share. A summary of the innovation driver of the money service provider for unbanked people business model is shown in Table 14.

Innovation driver		
Innovation opportunity	Market	
Competitive innovation dimension	Disruptive	
Newness innovation dimension	Revolutionary	

Table 14: Innovation driver of the money service provider for unbanked people business model

Next, we describe two examples of the money service provider for unbanked people business model.

6.6.1 Example 1: M-pesa

M-pesa stands for mobile cash (pesa is the Swahili word for cash). The Mpesa service is provided by he largest mobile phone operator in Kenya called Safaricom, part of the Vodafone group. M-pesa customers can use their mobile phones to transfer money between mobile phone users. The market targeted by Safaricom is the people without access to financial services. The customers do not need to have a bank account, they just need to register with Safaricom for a M-pesa account [67].

The M-pesa network has reached mored than 6.5 million customers in two years. The average daily transaction of the whole M-pesa network is 1.96 million dollars with an average of 20 dollars per transaction. The cost per transaction is much cheaper and convenient than the competition in Kenya. For example, to send 25 dollars by Western Union the charge is 57.6 % plus the 5 % for the post office. In contrast, M-pesa charges 2.8 percentage for each transaction [66].

6.6.2 Example 2: G-cash

G-cash brings payments and money transfers services by mobile phones in three predominant low-income rural provinces in the Philippines targeting over 100.000 people. This project aims to reduce the physical time consuming way to send money by trucks with a more convenient and secure way [68].

G-cash project is leaded by GXI inc., a subsidiary of Globe Telecom, which already has opened over 120,000 new G-cash accounts which have been used to send money domestically among other payment services [68].

6.7 Summary of the analysis

In Figure 12, we position the analysis of innovation-driven business models contextualized in the innovation pentagram from Section 3.2. Each business model is represented by a circle labeled with a number.



Figure 12: Emerging innovations visualization on the innovation pentagram for the financial industry

Business models innovations are P2P lending (labeled by the number "1") and social investing portfolio (labeled by the number "2"). These business model innovations are based on breakthrough innovations that are disrupting the market. On-line personal finance (labeled by the number "3") and electronic payment platform (labeled by the number "4") business models that are driven by channel innovation, which are based on breakthrough innovations, defining a new category in the market. The industrial equipment with telematics services business model (labeled by the number "5") is driven by experience innovation. In this case, the customer experience is enhanced by breakthrough innovation. The money service provider for unbanked people (labeled by the number "6") business model is driven by market innovation, using revolutionary innovation to disrupt the market. Business models driven by product innovation opportunity are not present in the analysis, because represents how incumbents commonly sustain their business models.

7 Related work

This work integrates innovation and business models frameworks to provide a method to analyze innovation-driven business models that are emerging in the financial services industry.

Most of the literature in financial services innovations focuses on the technological aspect. In [23], the innovation theory sustaining and disruptive innovations is depicted, but only exemplifies the electronic banking case without describing a general method to study innovations and does not address the business perspective.

[23] describes an extensive coverage of technological advances in financial services. The book covers technological aspects that are being currently used by traditional financial services companies and does not cover new entrants to this field. In [69], M.Shahrokhi presents an extensive literature survey on technological innovations in financial services. The paper lacks any kind of innovation and business model concepts. However, it offers a complete overview of electronic innovations in traditional financial services.

In [2], Daniel Fasnacht describes the benefits of open innovation in the financial services. It is a good guide to innovation, describing methods on how to embrace open innovation [2]. In [13], Gardner provides a framework of innovation opportunities in financial services based on innovation concepts and theories. In this report, we extend the innovations opportunities identified by Gardner to depict an analysis method for innovation-driven business models.

In [27], innovation opportunities for financial services are identified, but the innovation concepts and theory are missing. The inclusion of the innovation theory is important, because they don not address if the identified opportunity will sustain of change the business model of a company

In [13], Gardner identify innovation opportunities in the financial services to provide an innovation framework for the industry. We extended the innovation opportunities described on the book, incorporating business models frameworks to provide a structured method to analyze innovation-driven business models.

8 Conclusions

This report is a contribution to the innovation and business models fields for financial services, describing a unified method to analyze business models driven by innovations. Financial services companies can use this method to derive an overview of the state of the art and as a strategic tool to plan innovation initiatives.

In this report, we have described different innovation opportunities in the financial services industry. These opportunities are reached through the newness dimension of innovation, in which we can distinguish three kinds of innovations: breakthrough, revolutionary and incremental. Breakthrough innovations, are concerned with the exploration of new technologies. Revolutionary innovations are superior to what replace, becoming a standard choice for a relevant market share. Incremental innovations are minor changes, exploiting existing technology.

From the competitive position of the firm (competitive dimension), innovations can be disruptive or sustaining. Disruptive innovations are focused on simplicity, affordability or accessibility. The disruption of the market is made targeting new or less demanding customers. Sustaining innovations, increase the capabilities of the current offer, making more appealing the value proposition to the more demanding customers.

In this report, we have analyzed six emerging innovation-driven business models. Five business models are from insurgents and one is from incumbents. The proportion is explained by the less risk-averse culture by insurgents which are more prone to disrupt the market.

Business model innovation is the most disruptive kind of innovation opportunity. Creating new business models is not an easy task to incumbents, due to the conflict of interest with their currently successful business model. Incumbents may face the choice of disrupting the market or being disrupted by insurgents. Being aware of this is the first step to explore new business models that are often postponed due to operational priorities.

Innovating in business models is most natural to achieve through breakthrough innovation. However, breakthroughs innovations require to take risk in a risk-adverse industry like financial services. This duality is the cause of the low level of attention to the breakthrough innovations in established companies in the financial industry. Insurgents are leading business model innovation. Examples driven by breakthrough innovations are P2P lending and social investing portfolio. The exploration of new technologies is a convenient tool to disrupt the market, but it will not have an impact if these are not applied in an innovative business model.

Incumbents are using incremental innovations to innovate in channels to sustain their business models. In the meantime, insurgents are innovating in channels using breakthrough innovations to disrupt the market. Examples for channel innovation made by insurgents, are the on-line personal finance and electronic payment platform business models.

Presently, incumbents master product and market innovation to sustain their business model using incremental innovations. Products are delivered with more features and markets segments are reached without modifying the core product and business of the company. These incremental innovations do not offer the changes needed to disrupt the market. Insurgents, are using revolutionary innovation to provide specifically oriented solutions to new markets. An example for this kind of innovation is the money service provider for unbanked people which is specially designed for people without access to traditional bank accounts.

The main conclusion for the financial industry is that incumbents which only focus on improvement have limited options for future growth. A focus on breakthrough innovations is essential for incumbents to explore disruptive innovation. Companies must balance the exploitation of the current business with the exploration of new opportunities. However, this balance is hard to achieve due to the risk-averse culture in financial services. Chief executives need to position the exploration of breakthrough innovations as an important activity on financial services companies.

References

- [1] OECD, "OECD in figures 2008," tech. rep., OECD Publications, Paris, France, 2008.
- [2] D. Fasnacht, Open Innovation in the Financial Services. Springer, 1st ed., 2009.
- [3] T. L. Friedman, "The great disruption." http://www.nytimes.com/2009/ 03/08/opinion/08iht-edfriedman.1.20672274.html, March 2009. Last accessed on November 14, 2009.
- [4] S. Anthony, "The great disruption." http://blogs.harvardbusiness. org/anthony/2008/12/the_great_disruption.html, December 2008. Last accessed on October 17, 2009.
- [5] S. G. Shanker Ramamurthy and C. Petit, "Fit, focused and ready to fight," tech. rep., IBM Institute for Business Value, December 2009.
- [6] I. Walter, "Economic drivers of structural change in the global financial services industry," *Long Range Planning*, vol. 42, no. 5-6, pp. 588 – 613, 2009. Internationalization in Financial Services.
- [7] A. Osterwalder and I. Pigneur, *Business model generation*. self-published, 2009.
- [8] M. Venzin, Building an International Financial Services Firm: How Successful Firms Design and Execute Cross-Border Strategies in an Uneven World. Oxford University Press, USA, March 2009.
- [9] A. Kumar, Access to Financial Services in Brazil (Directions in Development). World Bank Publications, November 2004.
- [10] J. Surowiecki, "Why banks stay big." The New Yorker, November 2009.
- [11] P. K. Nevitt and F. J. Fabozzi, *Equipment Leasing*. Wiley, 4th edition ed., January 2000.
- [12] W. E. Forum, "The future of the global financial system," tech. rep., 2009.
- [13] J. Gardner, Innovation and the Future Proof Bank: A Practical Guide to Doing Different Business-as-Usual. Wiley, 1st ed., October 2009.
- [14] N. Haugen, "The informal credit market: A study of default and informal lending in nepal," Master's thesis, University of Bergen, 2005.
- [15] R. Hayler and M. Nichols, Six Sigma for Financial Services: How Leading Companies Are Driving Results Using Lean, Six Sigma, and Process Management. McGraw-Hill, 1st ed., December 2006.
- [16] CBI, "The shape of business the next ten years." http://www.cbi.org.uk, November 2009. Last accessed on January 31, 2010.
- [17] "Alexa top sites." http://www.alexa.com/topsites. Last accessed on March 10, 2010.

- [18] IBM, "The enterprise of the future in the banking industry," October 2008.
- [19] V. Wallace and A. Herrick, "What banking needs to become," *Strat-egy+business*, vol. Winter 2009, pp. 1–12, November 2009.
- [20] R. Landau, *Technology and Economic Policy*. Ballinger Pub Co, November 1986.
- [21] A. Kates and J. R. Galbraith, Designing Your Organization: Using the STAR Model to Solve 5 Critical Design Challenges. Jossey-Bass, November 2007.
- [22] C. M. Christensen and M. E. Raynor, The Innovator's Solution: Creating and Sustaining Successful Growth. Harvard Business Press, 1st ed., September 2003.
- [23] T. J. Albert Enders, Andreas König and H. Hubgenberg, "The relativity of disruption on the financial industry : e-banking as a sustaining innovation in the banking industry," *Journal of electronic commerce research*, vol. 7, no. 2, pp. 68–77, 2006.
- [24] S. D. Anthony, *The Silver Lining: An Innovation Playbook for Uncertain Times.* Harvard Business Press, 2009.
- [25] C. M. Christensen, *The Innovator's Dilemma*. Perseus Distribution Services, June 1997.
- [26] L. A. Costanzo and J. K. Ashton, "Product innovation and consumer choice in the u.k financial services industry," *Journal of Financial Regulation and Compliance*, vol. 14, pp. 285–303, July 2006.
- [27] J. Riberio, "Glittering prize glittering prize; how financial institutions can drive growth," tech. rep., Delloite Research, 2005.
- [28] S. L. Vargo and R. F. Lusch, "Evolving to a new dominant logic for marketing," *Journal of Marketing*, vol. 68, no. 1, pp. 1–17, 2004.
- [29] B. Bund, The Outside-In Corporation: How to Build a Customer-centric Organization for Breakthrough Results. McGraw-Hill, 1st ed., October 2005.
- [30] Y. J. Wind, "The challenge of customerization in financial services," Commun. ACM, vol. 44, no. 6, pp. 39–44, 2001.
- [31] Microsoft, "What is microsoft surface?." http://www.microsoft.com/ surface/Pages/Product/WhatIs.aspx. Last accessed on December 12, 2009.
- [32] C. Linke, "Logica and rabobank trail blaze with innovative interactive conference table." http://www.logica.com, April 2009. Last accessed on January 21, 2010.
- [33] R. Lommers, "Logica en rabobank starten proef met vergadertafel ms surface." http://www.raboblog.nl, March 2009. Last accessed on January 19, 2010.

- [34] "iDeal." http://www.ideal.nl. Last accessed on October 19, 2009.
- [35] G. Hamel, *Leading the Revolution: How to Thrive in Turbulent Times by Making Innovation a Way of Life.* Harvard Business Press, revised edition ed., October 2002.
- [36] D. J. Teece, "Business models, business strategy and innovation," *Long Range Planning*, vol. In Press, Corrected Proof, 2009.
- [37] C. M. Christensen, J. H. G. M.D., and J. H. M.D., *The Innovator's Prescription: A Disruptive Solution for Health Care.* McGraw-Hill, 1st ed., December 2008.
- [38] R. Amit and C. Zott, "Value creation in e-business," *Strategic Management Journal*, vol. 22, no. 6-7, pp. 493–520, 2001.
- [39] H. Chesbrough, "Business model innovation: Opportunities and barriers," *Long Range Planning*, August 2009.
- [40] A. Osterwalder, *The Business Model Ontology: A Proposition in a Design Science Aproach.* PhD thesis, Universite de Lausanne, 2004.
- [41] "ICT and e-business impact studies," tech. rep., The Sectoral e-Business Watch, 2009.
- [42] P. Grefen, Mastering e-Business. Routledge, 1st ed., May 2010.
- [43] "E-finance lab." http://www.efinancelab.de. Last accessed on September 13, 2009.
- [44] D. Tapscott, D. Ticoll, and A. Lowy, *Digital Capital: Harnessing the Power* of Business Webs. Harvard Business Press, 1st ed., May 2000.
- [45] A. G. Ning Chen and N. Lambert, "Social lending," EC '09: Proceedings of the tenth ACM conference on Electronic commerce, July 2009.
- [46] S. C. Berger and F. Gleisner, "Emergence of financial intermediaries in electronic markets: The case of online p2p lending," Official Open Access Journal of VHB, vol. 2, pp. 1–27, May 2009.
- [47] J. Sviokla, "Forget citibank borrow from Bob," February 2009. Last accessed on August 20, 2009.
- [48] M. Kupp and J. Anderson, "Zopa: Web 2.0 meets retail banking," *Business Strategy Review*, vol. 18, no. 3, pp. 11–17, 2007.
- [49] D. Roth, "Road map for financial recovery: Radical transparency now!." Wired Magazine, March 2009.
- [50] M. Palmer, "Covestor plans internet revolution." http://www.ft.com/ cms/s/0/ee3caa5e-a3b6-11de-9fed-00144feabdc0.html, September 2009. Last accessed on December 15, 2009.
- [51] J. Kim, "Web sites wade into money management." http://blogs.wsj. com/wallet/2009/07/29/web-sites-wade-into-money-management/, July 2009. Last accessed on December 15, 2009.

- [52] P. Crosman, "Kaching adds professional advice and live trading to its social network." http://www.wallstreetandtech.com/showArticle. jhtml?articleID=220601005, October 2009. Last accessed on November 13, 2009.
- [53] KaChing.com, "Genius frequently asked questions." http://www. kaching.com/support/faq/genius. Last accessed on December 12, 2009.
- [54] C. C. Miller, "Site lets investors see and copy experts trades." http://www. nytimes.com/2009/10/19/technology/start-ups/19kaching.html, October 2009. Last accessed on November 21, 2009.
- [55] S. B. Cohen and D. Schehr, "Microsoft bows to growth of online fsns in closing money plus," tech. rep., Gartner Research, June 2009.
- [56] W. E. Forum, "The world economic forum's technology pioneers 2009," December 2009.
- [57] D. Chartier, "Hands on: Wesabe helps you pinch pennies, avoid bank fees." http://arstechnica.com/web/news/2009/01/ hands-on-wesabe-helps-you-pinch-pennies-avoid-banke-fees. ars, January 2009. Last accessed on November 15, 2009.
- [58] "Wesabe springboard." https://www.getspringboard.com, 2009. Last accessed on December 16, 2010.
- [59] Amazon.com, "Amazon flexible payments service (amazon fps)." https: //payments.amazon.com, 2009.
- [60] M. Krueger and K. Leibold, "Internet payments in germany," *Handbook on Information Technology in Finance*, pp. 239–256, 2008.
- [61] PayPal, "Paypal adaptive payments developer guide." https://www.x. com, October 2009. Last accessed on January 19, 2010.
- [62] K. Phelan, "Using telematics to make money." http://www.cedmag.com/ article-detail.cfm?id=10924830, January 2009. Last accessed on January 20, 2010.
- [63] K. Hall, "How komatsu innovations keep its machinery selling." http://www.businessweek.com/print/globalbiz/content/sep2009/ gb20090930_232338.htm, September 2009. Last accessed on October 18, 2009.
- [64] T. Sumi, "Co-existing environment and economy of manufacturing industry in japan," Portland International Conference on Management of Engineering and Technology, pp. 2349–2355, August 20009.
- [65] VOLVO, "Volvo caretrack." http://www.volvo.com. Last accessed on January 16, 2010.
- [66] O. Lambert and E. Littlefield, "Dial growth," *Finance and development*, vol. 46, pp. 48–50, September 2009.

- [67] N. Hughes and S. Lonie, "M-pesa: Mobile money for the "unbanked", turning cellphones into 24-hour tellers in kenya," *Innovations: Technology, Governance, Globalization,* vol. 2, pp. 63–81, June 2007.
- [68] CGAP, "Gcash." http://www.cgap.org. Last accessed on December 21, 2009.
- [69] M. Shahrokhi, "E-finance: status, innovations, resources and future challenges," *International Journal of Managerial Finance*, vol. 34, no. 6, 2008.